Author's response to reviews

Title: Linkage of health and aged care service events: comparing linkage and event selection methods

Authors:

Rosemary Karmel (rosemary.karmel@aihw.gov.au)
Diana L Rosman (Diana.Rosman@health.wa.gov.au)

Version: 6 Date: 7 July 2008

Author's response to reviews:

BioMed Central Editorial Production Team

Requested formatting changes

Major revisions: The first three revision points have been made as requested. In particular, Table 4 has become Figure 1 and all table numbers and figure numbers have been adjusted as necessary. With respect to the fourth point, we feel that it is important to maintain the current table order, and so the reference to old Table 5 (new Table 4) in the first paragraph under the heading “Accuracy of E linkage” has been removed.

Minor revisions: Revisions have been made as requested.

A small number of changes to text relating to grammar or clarity of expression have also been made.

Please let me know if there are any necessary revisions remaining outstanding.

Yours sincerely

Rosemary Karmel